---
title: "R-Series: Reports"
description: "System status reports and performance analyses"
---

import { LinkCard, CardGrid, Badge } from '@astrojs/starlight/components';

System status reports and performance analyses

## Documents (19 total)

| Document | Description | Version | Date |
|----------|-------------|---------|------|
| [R006](/cospas-sarsat/r-series/r006/) | Document R006 |  |  |
| [R007](/cospas-sarsat/r-series/r007/) | No.41 Jan-Dec 2024 |  | December 2024 |
| [R009](/cospas-sarsat/r-series/r009/) | Summary Report Of The |  | October 1999 |
| [R010](/cospas-sarsat/r-series/r010/) | C/S R.010 - Issue 1 Rev. 8 | Issue 1 | October 2009 |
| [R011](/cospas-sarsat/r-series/r011/) | C/S R.011 Issue 1 Rev 1 | Issue 1 | October 2003 |
| [R012](/cospas-sarsat/r-series/r012/) | C/S R.012 - Issue 1 Rev 17 | Issue 1 | April 2023 |
| [R013](/cospas-sarsat/r-series/r013/) | Cospas-Sarsat Meteosat Second Generation | Issue 1 | October 2006 |
| [R014](/cospas-sarsat/r-series/r014/) | C/S R.014 |  | October 2009 |
| [R015](/cospas-sarsat/r-series/r015/) | R.015 | Issue 1 | October 2009 |
| [R016](/cospas-sarsat/r-series/r016/) | R16_28oct.2010 | Issue 1 | October 2011 |
| [R017](/cospas-sarsat/r-series/r017/) | C/S R.017 - Issue 1 Rev. 7 | Issue 1 | March 2021 |
| [R018](/cospas-sarsat/r-series/r018/) | Cospas-Sarsat Demonstration And Evaluation Plan For The | Issue 2 | February 2018 |
| [R019](/cospas-sarsat/r-series/r019/) | Cospas-Sarsat Electro Geosar Performance Evaluation Report C |  | October 2012 |
| [R020](/cospas-sarsat/r-series/r020/) | C/S R.020 Issue 1 |  | October 2012 |
| [R021](/cospas-sarsat/r-series/r021/) | Cospas-Sarsat Meosar System Demonstration And Evaluation Phase I Report C | Issue 2 | December 2015 |
| [R022](/cospas-sarsat/r-series/r022/) | C/S R.022 Issue 1 |  | February 2018 |
| [R023](/cospas-sarsat/r-series/r023/) | C/S R.023 - Issue 1 |  | February 2018 |
| [R024](/cospas-sarsat/r-series/r024/) | C/S R.024 - Issue 1 |  | March 2021 |
| [R025](/cospas-sarsat/r-series/r025/) | Draft Prelim. Issue A |  | October 2025 |

